Author's response to reviews

Title: Maternal exposure to polychlorinated biphenyls and the secondary sex ratio: an occupational cohort study

Authors:

  Carissa M Rocheleau (CRocheleau@cdc.gov)
  Stephen J Bertke (SBerke@cdc.gov)
  James A Deddens (JDDeddens@cdc.gov)
  Avima M Ruder (ARuder@cdc.gov)
  Christina C Lawson (CLawson@cdc.gov)
  Martha A Waters (MWaters@cdc.gov)
  Nancy B Hopf (Nancy.Hopf@hospvd.ch)
  Margaret A Riggs (MRiggs@cdc.gov)
  Elizabeth A Whelan (EWhelan@cdc.gov)

Version: 6 Date: 9 March 2011

Author's response to reviews:

Dear Editorial Production Team,

The attached version corrects several small typographical errors.

1) Page 2, the 'W' in EWhelan@cdc.gov is capitalized.

2) Page 9, paragraph 3, 'product' is rewritten as 'products'

3) Page 18, paragraph 2, sentence 1, "...detect a true relationship if PCBs- but only if PCBs...' is corrected to read, "...detect a true relationship- but only if PCBs..."

4) Page 21, reference 21, corrected "Substabcse" to "Substances"

5) Pages 29-31, tables 1-3: line weights were not consistent (only viewable in print or zoomed-in). These were corrected to be consistent with Table 4 (1.5 before/after title and at bottom of table, 0.75 pt internally).

6) Page 7, paragraph 2 should read "... and had given birth to a live infant (2,639 women)." instead of "and had given birth to a live infant after 7 months gestation (2,595 women)." Our exclusion of 44 additional women (yielding an analytic set of 2595 women) are described later in the methods.

Thank you,

Carissa Rocheleau